MINUTES OF FORWARD UK-V CLUSTER MEETING

DATE:

Tuesday 15th July 2003

TIME:

11.am ( Coffee from 10.30 a.m. )

VENUE
            Conference Room CR12




Rutherford Appleton Laboratory
1)      Minutes from Glasgow meeting


Agreed.

2)     Actions from the Glasgow meeting


All contained in the agenda

3)     Detector Update



Alex Chilingarov 
10min

Alex reported on two issues:-

 a) The classification of possible problem detectors found at Lancaster and Sheffield was reported. The classification types have been defined in earlier meetings. For the final detectors the numbers were as follows:-

 Electrically Weak
Reduced Quality
Warning List


15


46


61

Steve raised the question as to how these could be simply identified and removed (in the first instance) to leave top quality detectors for module construction. A discussion followed.

b) The effect of humidity on the HV characteristic was described. Although there were variations in individual cases that were not understood there seemed to be a clear trend in the variation of the shape of the HV characteristic with humidity. The I/V curves become flatter at high bias values as the humidity increases. It is important to check humidity when taking bias current values. The effect is thought to be caused by surface charging. This means that repeatability will depend on the humidity. 

4)      Status of Hybrid Acceptance Testing


Jeff Bizzell
10 min

Jeff showed a table giving the status of all hybrids received at RAL. In particular 1/3 of the K5.2 hybrids have been returned to Hybrid SA. The problems are mainly due to poor bonding caused by contamination on the bond pads + poor QA at Hybrid SA.

Overall there has been a slip of 12 weeks in the hybrid production.

Of the 4 hybrids available one will go to Valencia.

The DORIC chip has a problem 5% of the time when it produces unwanted clock signals on the redundancy link.

It is hope d that in production the burn-in can be reduced from 90 hrs to 60 hrs.

ACTION
Ashley to consider how to test for this

5)      Reports on  Module Qualification, recent results   




Manchester







10 min
Steve reported on the recent production of 1 mechanical module and 2 qualifiaction modules. Modifications have been made so that the metrology of modules following M O 048 all show an improvement in precision. The SmartScope is now fully operational. 

There is still a 2 micron offset between Liverpool measurements and others which is thought to be due to a tilt in the thick glass fiducials. Pinhole fiducials have been prepared and Steve will test them out at Liverpool this week.

ACTION 
Report on result

Spine 69 was out of spec. and had a crack in the TPG

Spine 36 needed pieces of TPG trimming off it


Liverpool







10 min

Liverpool had a 5th module which was undergoing long term testing. An earlier module needed to have the testing completed by performing an I/V to 500V. The data are available on the Web and will soon be added to the database.
 

ACTION Continue Qualification procedure


Glasgow







10 min

There were reports on the two pre-qualification modules. 

MO 040 showed a slight oscillation at about 0.6 fC

MO 044 had no oscillations visible and passed the I/V test.

ACTION Continue Pre-qualification procedure

Valencia







10 min

There were reports on 5 pre-qualification modules – 2 mechanical and 3 electrical.

The last 3 showed bad planarity. Some variations ~ 4-8m were found after thermal cycling. I\V was OK. Bond heights and the module envelope were measured. Some bond heights were out of tolerance. There were 2-4 unbonded channels per module.

Modules were exchanged with Liverpool to check metrology. A difference of 2 m was found.

Three electrical modules were within specification but show a distortion at 0.3 fC.

Preparations for the production of middle modules are underway.

ACTION Continue Pre-qualification procedure
6)      Outstanding QA Issues



Joe Foster
10 min

          Discussion of sample tests and rework

7)      Procurement schedule for module components,


         staffing,and production plans for the next two months


      Phil Allport    10 min
The CS cluster was close to qualification..  Geneva is pretty well there. CERN has some more work to do. Melbourne is just starting. Cracow were having difficulties using the AERO.

ACTION
Ashley to help

In the CE cluster NIKHEF are about ready. Friburg and MPI are in the preparation stage.

Phil listed the available components, particularly hybrids and spines. For the hybrids several were sent back to hybrid SA with faults, mainly bonding problems.

The AlN will be at CERN by 1/8/2003. Phil showed the latest schedule for the spines.

In general the schedule is looking tighter than ever. See transparencies for details.

ACTION 
Craig to get Paul Dervan to check deliveries in and out of CERN for the Spines.


8)
Status of the irradiation programme

Craig Buttar
10 mins

The annual service of T7 was done in January. The new rotary target station has been installed and commissioned. A new camera is installed and is working well and the new laboratory is ready. On the baisis of one module it was found that the high gain spread seen on the ABCDs seemed to go away after irradiation.

See the transparencies for pictures and results.

9)
Time constant for decay of inter-strip capacitance

             at low humidity   




Dinos  Bachas 10 mins

Ian reported on some studies made by his student Dinos. The interstrip capacitance was measured by bonding 5 adjacent strips together and measuring the capacitance with a HP LCR at 10kHz, 100kHz and 1 MHz.. The time constant varied from 19 min. at a humididty of 29.8% to 218 min. at a humidity of 4.7%. For the latter point the current fell to a minimum after ~1 hr and then increased linearly.

10)         Common Mode Noise for forward modules at system test








Irina Nasteva

10 min

Ian reported on some work done by Irina at the Systems Test. The idea was to measure common mode noise. At the systems test this is injected artificially. Various software methods were used including looking at the number of hits per event and  the  and  methods as previously used by the collaboration. The studies are at an early stage and will continue.

ACTION
Report at next UK-V Meeting

11)
Status of the System Test


 Richard Bates 
10 min

Work continues to upgrade the systems test. 6 of Val’s new tapes have been installed on the disc. The ends of the tapes have been reinforced. Jumpers have been added to PPF0 and the grounding foils have been added. The system can cope with either SAMTEC or SPECIALITY connectors. It is estimated that all will be complete in a week or so.

